Author's response to reviews

Title: Metformin efficacy and safety for colorectal polyps: A double-blind randomized controlled trial

Authors:

Takuma Higurashi (takumahigurashi@yahoo.co.jp)
Hirokazu Takahashi (hirokazu@med.yokohama-cu.ac.jp)
Hiroki Endo (endo1978@yokohama-cu.ac.jp)
Kunihoro Hosono (hiro1017@fukuhp.yokohama-cu.ac.jp)
Eiji Yamada (eiji@yokohama-cu.ac.jp)
Hidenori Ohkubo (ohkuboh@yokohama-cu.ac.jp)
Eiji Sakai (eiji-not_found@mail.goo.ne.jp)
Takashi Uchiyama (maruru0203@yahoo.co.jp)
Yasuo Hata (yhata-gi@umin.ac.jp)
Nobutaka Fujisawa (fujisawa-yokohama2640@hotmail.co.jp)
Shiori Uchiyama (shiori_u_0120@yahoo.co.jp)
Akiko Ezuka (aki_ezuka@yahoo.co.jp)
Hajime Nagase (nagase@yokohamah.rofuku.go.jp)
Takaomi Kessoku (takaomi-kesso@hotmail.co.jp)
Nobuyuki Matsuhashi (nmatuha-tky@umin.ac.jp)
Shoji Nakayama (takuma_h@yokohama-cu.ac.jp)
Yoshiaki Inayama (takuma_higu777ln_nl@yahoo.co.jp)
Satoshi Morita (smorita@urahp.yokohama-cu.ac.jp)
Atsushi Nakajima (nakajima-tyk@umin.ac.jp)

Version: 6 Date: 14 March 2012

Author's response to reviews: see over
To Christna Chap, PhD
Senior Executive Editor
BMC Cancer

MS: 1791245421654997

Metformin efficacy and safety for colorectal polyps: A double-blind randomized controlled trial
by Takuma Higurashi

Dear Christna

Thank you for your letter.

We submitted revised manuscript that included author’s contribution at 23th Jan. However we mistook and submitted previous version. I apologize you for my mistake. In revise version, we described Author’s contribution in page 12 and 13.

Metformin is a biguanide derivative that has been long and widely used for treating diabetes mellitus. Recent study reported that patients with type2 diabetes taking metformin might be at a lower risk of cancer (including CRC) compared with those patients who do not take metformin. We conducted a prospective randomized control trial to evaluate the chemopreventive effects and drug safety of metformin against metachronous colorectal polyps in nondiabetic post-polypectomy patients. We believe
that many physicians will be interested in our research. We hope that you will find out our article suitable for your esteemed journal.

Yours sincerely,

Takuma Higurahi, MD.
Atsushi Nakajima, MD.

Division of Gastroenterology, Yokohama City University School of Medicine.
3-9 Fuku-ura, Kanazawa-ku, Yokohama 236-0004 Japan

Phone: +81-45-7872640    Fax: +81-45-7843546

E-mail: nakajima-ty@umin.ac.jp